Author's response to reviews

Title: Gender-based differences in the high-risk sexual behaviours of young people in Melilla (Spain): a transversal study

Authors:

Esperanza Romero-Estudillo (mariaesperanzaromero@ugr.es)
Emilio González-Jiménez (emigoji@ugr.es)
María C. Mesa-Franco (mcmesa@ugr.es)
Inmaculada García-García (igarcia@ugr.es)

Version: 2
Date: 26 February 2014

Author's response to reviews: see over
Response to Reviewers:

1) Research involving human subjects, human material, or human data, must have been performed in accordance with the Declaration of Helsinki (http://www.wma.net/en/30publications/10policies/b3/index.html) and must have been approved by an appropriate ethics committee. Experimental research on vertebrates or any regulated invertebrates must comply with institutional, national, or international guidelines, and where available should have been approved by an appropriate ethics committee. A statement detailing what ethical approval has been obtained, including the name of the ethics committee and the reference number where appropriate, must appear in all manuscripts. If a study has been granted an exemption from requiring ethics approval or does not require approval, this should also be detailed in the manuscript (including the name of the ethics committee that granted the exemption). For all research involving human subjects, informed consent to participate in the study should be obtained from participants (or their parent or guardian in the case of children under 16) and a statement to this effect should appear in the manuscript. Please refer to our Editorial Policies for full details of the ethical requirements for the journal: http://www.biomedcentral.com/about/editorialpolicies#Ethics.

This information has been incorporated in the manuscript.

2) By way of a section ‘Acknowledgements’, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible. The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as ‘We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.’ Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

This information has been incorporated in the manuscript.
3) You have uploaded the tables as figure files. Please remove them from the submission system and include the tables within the text file of the manuscript after the references. The tables should be formatted using the Table tool in your word processor. Please also move the table title to above the table and the legend to below the table, within the text.

The tables have been included after the references